Author’s response to reviews

Title: The use of MRI apparent diffusion coefficient (ADC) in monitoring the development of brain infarction

Authors:

Jianmin Shen (Jianmin_shen@163.com)
XIanwu Xia (XIanwu_xia@163.com)
Wugen Kang (Wugen_kang@163.com)
Jianjun Yuan (Jianjun_yuanyuan@163.com)
Liang Sheng (Jianjun_yuanyuan@163.com)

Version: 2 Date: 24 August 2010

Author’s response to reviews:

Revision made:

- Ethics: The ethics approval was obtained by Taizhou Medical college committee of clinical research, and written form approvals were obtained from patients.

- authors’ contributions: modified.

- misspelling of infarction as infraction: corrected

- abstract: formatted.